Author's response to reviews

Title: Retrospective comparison between a regular and a split-dose protocol of 5-fluorouracil, cisplatin, and mitoxantrone for the treatment of far advanced hepatocellular carcinoma

Authors:

Chau-Ting Yeh (chautingy@gmail.com)
Hui-Chin Chen (huichin11@adm.cgmh.org.tw)
Chang-Mu Sung (andrew3329@adm.cgmh.org.tw)
Cheng-Lung Hsu (hemaonco@admin.cgmh.org.tw)
Chen-Chun Lin (lincc53@adm.cgmh.org.tw)
Kuang-Tse Pan (pan0803@adm.cgmh.org.tw)
Jeng-Hwei Tseng (ma0389@adm.cgmh.org.tw)
Chien-Fu Hung (cfhung@adm.cgmh.org.tw)

Version: 5 Date: 24 March 2011

Author's response to reviews: see over
Dear Editor,

Thank you for the suggestions of formatting our paper. We have made the requested changes accordingly.

Major:

A short title was given for the figure.

Minor:

All text was changed to black.

Visible vertical lines in Tables were removed.

Typographical errors were checked.

Thank you again for handling our paper.

Sincerely,

Chau-Ting Yeh